Author's response to reviews

Title: Activity and Life After Survival of a Cardiac Arrest (ALASCA) and the effectiveness of an early intervention service: design of a randomised controlled trial [ISRCTN74835019]

Authors:

Veronique RMP Moulaert (v.moulaert@srl.nl)
Jeanine A Verbunt (j.verbunt@srl.nl)
Caroline M van Heugten (c.vanheugten@np.unimaas.nl)
Wilbert GM Bakx (w.bakx@srl.nl)
Anton PM Gorgels (a.gorgels@cardio.azm.nl)
Sebastiaan CAM Bekkers (b.bekkers@cardio.azm.nl)
Marc CFTM de Krom (m.dekrom@neurologie.azm.nl)
Derick T Wade (derick.wade@dsl.pipex.com)

Version: 4 Date: 13 August 2007

Author's response to reviews:

Ms Norton
Editor in chief
BMC Cardiovascular Disorders

Subject: Submission of an original design article
Hoensbroek, August 13, 2007

Dear Ms Norton, editor in chief,

We are very pleased to hear that our manuscript with the title: 'Activity and Life After Survival of a Cardiac Arrest (ALASCA) and the effectiveness of an early intervention service: design of a randomised controlled trial [ISRCTN74835019]' has been accepted for publication as an original design article in your journal.

We have made the revisions as mentioned in the email, changed the files in the submission system accordingly and hope that you consider the manuscript ready for publication.

Yours sincerely,
Veronique R.M.P. Moulaert
Corresponding author:
Veronique RMP Moulaert, MD, trainee in rehabilitation medicine
Rehabilitation Foundation Limburg
P.O. box 88
6430 AB HOENSBROEK
The Netherlands
E-mail: v.moulaert@srl.nl